Author’s response to reviews

Title: Prevalence of HPV high and low risk types in cervical samples from the Italian general population: a population based study.

Authors:

Paolo Giorgi Rossi (giorgirossi@asplazio.it)
Simonetta Bisanzi (s.bisanzi@ispo.toscana.it)
Irene Paanini (i.paganinni@ispo.toscana.it)
Angela Di Iasi (moscati.anatpat@aslcaserta2.it)
Claudio Angeloni (claudio.angeloni@aslt eramo.it)
Aurora Scalisi (screegin@ausl3.ct.it)
Rosalba Macis (centrodonnacagliari@asl8cagliari.it)
Maria T Pini (matinf.dip59@asl2.napoli.it)
Francesco Chini (chini@asplazio.it)
Francesca M Carozzi (f.carozzi@ispo.toscana.it)
HPV prevalence Italian Working Group (camilloni@asplazio.it)

Version: 6 Date: 13 July 2010

Author’s response to reviews:

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution

OK, we reported the institution as suggested. Not all the institutions have a street address.

Author groups: please include the name of the group within the list of authors, by adding the name to the `first name box`; and the bracketed abbreviation to the `last name' box in the submission system.

Please remove the names of the members of the group from the title page and place them in the Acknowledgements, as they are to be acknowledged as members of the study group.

The groups’ contribution as a whole can be given in the authors contributions section.

OK we moved the group components in the acknowledgement. Actually in the authors’ contribution the WG is already mentioned as a whole.

We also changed the way of filling the authors form.

Box: unfortunately we cannot incorporate `box 1', or text boxes. Please either
change the title to `Table 1', please remember to up-date any references to `Box 1' within the text, or include the information within the manuscript text- you can use indentation to highlight the text

OK we changed the box in a table.

References: Please make sure references are cited in numerical order.

OK: we had a problem with the table 5 references: now

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

OK

Introduction: please rename this section 'Background'.

OK

Materials and methods: please rename this section 'Methods'.

OK

Abbreviations: Please format these in to a paragraph.

Actually a paragraph reporting all the abbreviations was already present just before the Acknowledgement.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

We did not change the figures, we hope the figures were OK

References: the reference list should contain up to 30 authors' names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.

We corrected all the references.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

OK we hope there are not typing errors, but actually we could not make a further revision with an English speaking editor in so few days.